Author’s response to reviews

Title: Seasonal influenza vaccination coverage and its determinants among nursing homes personnel in Western France.

Authors:

Christelle ELIAS (christelle137@hotmail.com)
Anna FOURNIER (annafournier87@gmail.com)
Anca VASILIU (ancavasiliuro@yahoo.com)
Nicolas BEIX (nicobeix@gmail.com)
Rémi DEMILLAC (demillac@club-internet.fr)
Hélène TILLAUT (helene.tillaut@ars.sante.fr)
Yvonnick GUILLOIS (Yvonnick.GUILLOIS@ars.sante.fr)
Serge EYEBE (drsergeeyebe@netcourrier.com)
Bastien MOLLO (bastien.mollo@gmail.com)
Pascal CREPEY (crepey@gmail.com)

Version: 2 Date: 06 Mar 2017

Author’s response to reviews:

Editor Comments:

(1) Please include a 'Declarations' section heading after the Abbreviations.

(2) Please rename the ethics section 'Ethics approval and consent to participate' and state in this section whether you discussed your study with an ethics committee and whether you received an official ethics waiver. You should also provide the name of the ethics committee which granted this.

Response: We thank the reviewers and the editor for their acceptance of our manuscript. We have modified the manuscript according to the editor recommendations.